Author's response to reviews

Title: Adult Growth Hormone Deficiency (AGHD) treatment with a combination of Growth Hormone (GH) and Insulin-like Growth Factor 1 [IGF-1] (Increlex) resulting in elevated sustainable IGF-1 and IGF Binding Protein 3 (BP3) plasma levels: A case report

Authors:

Eric R Braverman (Pathmedical@aol.com)
Abdalla Bowirrat (bowirrat@netvision.net.il)
Uma J Damle (udamle04@yahoo.com)
Swetha Yeldandi (swethayeldandi@gmail.com)
Thomas J H Chen (tjhchen@yahoo.com.tw)
Margaret Madigan (MargBetts@hotmail.com)
Mallory Kerner (mallory.kerner@gmail.com)
Stanley X Huang (xhuang56@hotmail.com)
Stella Savarimuthu (stella.savarimuthu@gmail.com)
Kenneth Blum (drd2gene@aol.com)

Version: 3 Date: 27 April 2010

Author's response to reviews: see over
Dear Sir:

Please find our revised manuscript by Braverman et al. “Adult Growth Hormone Deficiency (AGHD) treatment with a combination of Growth Hormone (GH) and Insulin-like Growth Factor 1 [IGF-1] (Increlex) resulting in elevated sustainable IGF-1 and IGF Binding Protein 3 (BP3) plasma levels: A case report” for your review.

We have made the following changes, highlighted in yellow within the manuscript:

1) On page 5, added ethnicity of the subject
2) On page 8, updated the consent section to read, “Written informed consent was obtained from the patient for publication of this case report and accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal.”

Looking forward to a favorable review in a timely manner.

Professor Kenneth Blum, PhD.